Author’s response to reviews

Title: Do endometrial lesions require removal? A retrospective study

Authors:

Ting Jiang (cherryj2013@hotmail.com)
Qing Yuan (yuanqing@51mch.com)
Qin Zhou (qzhou@shmu.edu.cn)
Yiping Zhu (zhuyiping@51mch.com)
Siji Lv (lvsiji@51mch.com)
Yanling Cao (caoyanling@51mch.com)
Qin Wang (13122557949@163.com)
Kunming Li (likunming@hotmail.com)
Dong Zhao (hendryz@gmail.com)

Version: 5 Date: 10 Apr 2019

Author’s response to reviews:

Dear Editor:

We had made some formatting changes according to your requests.

1. Table 1 is cited within the second paragraph of the results section.

2. We have uploaded the Table 2.

3. We think we have uploaded the clean version. If we had misunderstand, please tell us where we should revise in details.

Thank you for your continued patience.

Sincerely,

Dong Zhao M.D.